# NANOMATERIALS: EVOLUTION AND ADVANCEMENT TOWARDS THERAPEUTIC DRUG DELIVERY (PART II)

Editors: Surendra Nimesh Nidhi Gupta Ramesh Chandra

**Bentham Books** 

# Nanomaterials: Evolution and Advancement towards Therapeutic Drug Delivery *(Part II)*

Edited by

# Surendra Nimesh

Department of Biotechnology Central University of Rajasthan India

# Nidhi Gupta

Department of Biotechnology IIS (Deemed to be University) India

# &

# **Ramesh Chandra**

Department of Chemistry University of Delhi India

## Nanomaterials: Evolution and Advancement towards Therapeutic Drug Delivery (Part II)

Editors: Surendra Nimesh, Nidhi Gupta and Ramesh Chandra

ISBN (Online): 978-1-68108-823-5

ISBN (Print): 978-1-68108-824-2

ISBN (Paperback): 978-1-68108-825-9

© 2021, Bentham Books imprint.

Published by Bentham Science Publishers - Sharjah, UAE. All Rights Reserved.

## BENTHAM SCIENCE PUBLISHERS LTD.

#### End User License Agreement (for non-institutional, personal use)

This is an agreement between you and Bentham Science Publishers Ltd. Please read this License Agreement carefully before using the ebook/echapter/ejournal (**"Work"**). Your use of the Work constitutes your agreement to the terms and conditions set forth in this License Agreement. If you do not agree to these terms and conditions then you should not use the Work.

Bentham Science Publishers agrees to grant you a non-exclusive, non-transferable limited license to use the Work subject to and in accordance with the following terms and conditions. This License Agreement is for non-library, personal use only. For a library / institutional / multi user license in respect of the Work, please contact: permission@benthamscience.net.

#### **Usage Rules:**

- 1. All rights reserved: The Work is the subject of copyright and Bentham Science Publishers either owns the Work (and the copyright in it) or is licensed to distribute the Work. You shall not copy, reproduce, modify, remove, delete, augment, add to, publish, transmit, sell, resell, create derivative works from, or in any way exploit the Work or make the Work available for others to do any of the same, in any form or by any means, in whole or in part, in each case without the prior written permission of Bentham Science Publishers, unless stated otherwise in this License Agreement.
- 2. You may download a copy of the Work on one occasion to one personal computer (including tablet, laptop, desktop, or other such devices). You may make one back-up copy of the Work to avoid losing it.
- 3. The unauthorised use or distribution of copyrighted or other proprietary content is illegal and could subject you to liability for substantial money damages. You will be liable for any damage resulting from your misuse of the Work or any violation of this License Agreement, including any infringement by you of copyrights or proprietary rights.

#### **Disclaimer:**

Bentham Science Publishers does not guarantee that the information in the Work is error-free, or warrant that it will meet your requirements or that access to the Work will be uninterrupted or error-free. The Work is provided "as is" without warranty of any kind, either express or implied or statutory, including, without limitation, implied warranties of merchantability and fitness for a particular purpose. The entire risk as to the results and performance of the Work is assumed by you. No responsibility is assumed by Bentham Science Publishers, its staff, editors and/or authors for any injury and/or damage to persons or property as a matter of products liability, negligence or otherwise, or from any use or operation of any methods, products instruction, advertisements or ideas contained in the Work.

### Limitation of Liability:

In no event will Bentham Science Publishers, its staff, editors and/or authors, be liable for any damages, including, without limitation, special, incidental and/or consequential damages and/or damages for lost data and/or profits arising out of (whether directly or indirectly) the use or inability to use the Work. The entire liability of Bentham Science Publishers shall be limited to the amount actually paid by you for the Work.

#### **General:**

- 1. Any dispute or claim arising out of or in connection with this License Agreement or the Work (including non-contractual disputes or claims) will be governed by and construed in accordance with the laws of the U.A.E. as applied in the Emirate of Dubai. Each party agrees that the courts of the Emirate of Dubai shall have exclusive jurisdiction to settle any dispute or claim arising out of or in connection with this License Agreement or the Work (including non-contractual disputes or claims).
- 2. Your rights under this License Agreement will automatically terminate without notice and without the

need for a court order if at any point you breach any terms of this License Agreement. In no event will any delay or failure by Bentham Science Publishers in enforcing your compliance with this License Agreement constitute a waiver of any of its rights.

3. You acknowledge that you have read this License Agreement, and agree to be bound by its terms and conditions. To the extent that any other terms and conditions presented on any website of Bentham Science Publishers conflict with, or are inconsistent with, the terms and conditions set out in this License Agreement, you acknowledge that the terms and conditions set out in this License Agreement shall prevail.

Bentham Science Publishers Ltd.

Executive Suite Y - 2 PO Box 7917, Saif Zone Sharjah, U.A.E. Email: subscriptions@benthamscience.net



## CONTENTS

FOREWORD	i
PREFACE	ii
LIST OF CONTRIBUTORS	iv
CHAPTER 1 SOLID LIPID NANOPARTICLES AND NANOSTRUCTURED LIPID	
CARRIERS FOR DRUG DELIVERY APPLICATIONS	1
Gabriel Silva Borges, Mariana Silva Oliveira, Délia Chaves Moreira dos Santos, Lucas	
Antônio Miranda Ferreira and Guilherme Carneiro	
INTRODUCTION	1
BACKGROUND	2
COMPOSITION OF SLN AND NLC	8
Lipids	
Surfactants	12
Other Components	15
METHODS FOR SLN AND NLC PRODUCTION	
Hot or Cold Homogenization Technique	16
High Pressure Homogenization (HPH)	
Ultrasonication	
O/W Microemulsion	
Emulsification-Solvent Evaporation	
Solvent Injection Method	18
Multiple Emulsion	19
Membrane Contactor Method	19
Super Critical Fluid Technique	19
Drug Distribution in Lipid Nanocarriers	20
Strategies for Improving Encapsulation of Drugs in Lipid Nanoparticles	
PHYSICOCHEMICAL CHARACTERIZATION	23
Particle Size	23
Dynamic Light Scattering	
Laser Diffraction	24
Scanning Ion Occlusion Spectroscopy	25
Zeta Potential	25
Crystallinity and Polymorphism	25
Differential Scanning Calorimetry	
X-ray Diffraction	
Particle Shape and Morphology	27
Drug Encapsulation Efficiency and Drug Loading	28
Soluble in Aqueous Medium	
Precipitated	
Total Drug Concentration	29
COMMON PROBLEMS IN SLN AND NLC	29
Gelation	29
Supercooled Melts	
Coexistence of Nanostructures	
MAIN APPLICATIONS OF SLN AND NLC IN PHARMACEUTICS	
Administration Routes	
Dermal Route	
Oral Route	
Pulmonary Route	35

Intravenous Administration	35
Intranasal Delivery	35
Ophthalmic Delivery	36
Important Applications of SLN and NLC	
Cancer	
Brain Delivery	
CONCLUSION	
CONSENT FOR PUBLICATION	
CONFLICT OF INTEREST	40
ACKNOWLEDGEMENTS	40
REFERENCES	40
CHAPTED 2 SHICA RASED NANOMATERIAL FOR DRUC DELIVERV	57
Charu Bharti Md Sabir Alam Md Nouchad Javad Muhammad Khalid Saifullah	
Eaisal A. Almalki and Pomila Manchanda.	
INTRODUCTION	57
CHEMISTRY OF MESOPODOUS SUICA NANOPARTICI ES	
CHADACTEDISTICS OF MENS	
CHADACTEDIZATION TECHNIOUES USED FOD MSNS	61
ET ID Spectra	01
Flatton Migroscom	01
METHODS OF DDFD A D A TION OF MSNS	
Stober Process	
Evanoration Induced Assembly Synthesis	
Combustion Synthesis Process	
Aerosal Decomposition Synthesis	
Chemical Vanour Deposition	
Photothermal Synthesis	
Inert Gas Condensation	
SURFACE FUNCTIONALIZATION OF SILICA NANOPARTICLES BV VARIO	
METHODS	68
Co-condensation Method	68
Post-synthetic Grafting	69
BIOCOMPATIBILITY AND BIO-DISTRIBUTION	69
Impact of Size and Shape of the Particle	70
Impact of Surface Properties	70
APPLICATIONS OF MSNS	71
MSNs as siRNA Delivery Vehicles for Cancer Therapy	
Gene Delivery / Drug Delivery Using MSNs	
Bio-medical Imaging	
The Imaging Capability of MSNs in Biomedical	
Controlled Release of Bioactive Molecules	
Green Chemistry Application of MSNs	
Protein Adsorption and Separation	
Vascular Targeting/ Passive Targeting	
PATENTED CITATION OF MESOPOROUS SILICA NANOPARTICLES	
CONSENT FOR PUBLICATION	
CONFLICT OF INTEREST	
ACKNOWLEDGEMENTS	81
REFERENCES	81

CHAPTER 3 HYDROGELS FOR DRUG DELIVERY	
Reena Singh, Rahul Goel, Seema Gupta and Pradeep Kumar	
INTRODUCTION	
POLYMER-BASED HYDROGELS	
Natural Polymer-based Hydrogels	
Chitosan-based Hydrogels	
Gelatin-based Hydrogels	
Hyaluronic Acid-Based Hydrogels	
Pullulan-Based Hydrogels	
An Alginate-Based Hydrogel	
Protein-Based Hydrogel	
Synthetic Polymer-Based Hydrogels	100
Polvethylene Glycol (PEG)-Based Hydrogels	100
Thermally Insulated Hydrogel	103
Polvlactic Acid (PLA)-Based Hydrogels	103
Polyvinyl Alcohol (PVA)-Based Hydrogel	104
Polyacrylic Acid-Based Hydrogels	104
Polyglutamic Acid-Based Hydrogels	105
Poloxamer Based Hydrogel	105
Polylysine-Based Hydrogel	106
PEPTIDE-BASED HYDROGELS	107
RADA-16-Based Hydrogels	107
R-Amino Acid Pentide-Based Hydrogel	111
Other Miscellaneous Polynentide Based Hydrogels	112
FUTURE PERSPECTIVES	117
CONSENT FOR PUBLICATION	117
CONFLICT OF INTEREST	117
ACKNOWLEDGEMENTS	117
REFERENCES	118
CHAPTER 4 METALLIC NANOPARTICLES: APPLICATIONS IN DRUG DELIVERY	125
Monisha Singhal, Nidhi Gupta and Jyoti Dasharath Magare	105
INTRODUCTION	
ADVANTAGES OF NANOPARTICLES-BASED DRUG DELIVERY	126
APPLICATIONS	127
METALLIC NANOPARTICLES	128
History	128
Characteristic Properties	129
GOLD NANOPARTICLES	129
Methods for the Synthesis of AuNPs	129
Chemical Method	130
Physical Methods	130
Green Methods	131
Biological Methods	131
Application of AuNPs	132
AuNPs as Drug Delivery on Various Diseases	133
AuNPs as Drug Delivery Vehicle for the Therapy of Metabolic Diseases	133
AuNPs as Drug Delivery Vehicles for Anti-infective Agents	133
AuNPs as Drug Delivery on Cancer	134
AuNPs as the Delivery of Gene	135
SILVER NANOPARTICLES	136
Methods for the Synthesis of AgNPs	136

Chemical Method	137
Physical Method	137
Green Method	137
Biological Method	138
Application of AgNPs	138
AgNPs for Drug Delivery in Various Diseases	139
AgNPs in Wound Healing	139
AgNPs Impregnated on Catheters	139
AgNPs as Antimicrobials	140
AgNPs as a Tool for Sensing Applications	140
METAL OXIDE NANOPARTICLES	
Zinc Oxide Nanoparticles	141
Iron Oxide Nanoparticles	
Titanium Oxide Nanoparticles	
Cerium Oxide Nanoparticles	142
Copper Oxide Nanoparticles	
Silicon Dioxide or Silica Nanoparticles	143
CONCLUSION	143
CONSENT FOR PUBLICATION	144
CONFLICT OF INTEREST	144
ACKNOWLEDGEMENTS	144
REFERENCES	145
CHAPTER 5 COMPUTATIONAL AND EXPERIMENTAL RINDING INTERACTION	SOF
CHAFTER 5 COMPUTATIONAL AND EAFENIMENTAL DINDING INTERACTION DDUC AND D CVCLODEVTDIN AS A DDUC DELIVEDV VEHICLE	5 OF 151
Chanduahoso Sahanai and Sanion Kuman Singh	131
INTRODUCTION	151
Chronological Evolution of Cyclodevtrin	
COMPLEX FORMATION IN CVCL ODEXTRIN	155
Dhysical Blanding	155
Slurry Complexation	
Kneeding Technique	150
Co-Precipitation Method	
Spray-Drving Method	
Microwaye Irradiation Method	
Milling or Co-Grinding Method	
Machanism of Drug Palaose from Cyclodeytrin Compley	
CVCLODEVTRIN BASED SUPRAMOLECULAR VESICLES	
SHAPF AND CHEMISTRY OF CVCI ODFYTRIN	162
PROPERTIES OF CVCI ODEXTRINS	
Computational findings of Drug. B. Cyclodevtrin Inclusion Complex	
Docking Studies on B cyclodextrin	105
Cyclodextrin Based Anticancer Drugs	
Cyclodextrin as Dharmaceutical Excinients	
Pharmacoutical Applications of Cyclodortrin	
Bioavailability Enhancement	170
Active Stabilization	1/1 172
Arrive Staulization	1/2
CVCI ODEVTRIN DERIVATIVES	1/2
UTUUUEATAIN DENIVATIVES Hydrophilio Darivativas	1/3
	172
Hydroxyallol Derivatives	

Sulfated Cvclodextrins	
Hydrophobic Derivatives	174
EXPERIMENTAL VERIFICATION IN DRUG-R- CYCLODEXTRIN INCLUSIO	N
COMPLEX	174
Thermal Analytical Technique	175
X-ray Diffraction	175
Spectroscopy Studies	176
Visible and Ultraviolet (Uv) Spectroscopy	176
Scanning Electron Microscone	176
DRUG RELEASE IN DRUG-CVCL ODEXTRIN	
SVNTHESIS OF CVCLODEXTRIN	178
PRODUCTION OF CYCLODEXTRIN	
Solvent Process	
SUCCESS OF RETA- CVCLODEXTRIN	
CHALLENGES AND FUTURE PERSPECTIVES	
CONSENT FOR PUBLICATION	
CONFLICT OF INTEREST	
ACKNOWI FDCFMENTS	
ACKNOW LEDGEMENTS	
CHAPTER 6 CLINICAL MILESTONES IN NANOTHERAPEUTICS: CURRENT ST	ATUS
AND FUTURE PROSPECTS	194
Ravi Goyal, Rupinder K. Sodhi and Jitender Madan	
INTRODUCTION	194
DRUG TARGETING	195
CARRIERS FOR NANOTHERAPEUTIC FORMULATIONS	195
Solid Lipids Nanoparticles	196
Polymeric Nanoparticles	197
Nanoliposomes	200
Micelles	200
Dendrimers	201
Nanosuspensions	201
Carbon Nanotubes	202
Fullerenes	203
Magnetic Nanoparticles	204
NANOTHERAPEUTICS IN CANCER	
Cancer Imaging	205
Chemotherapeutics	206
Nanoliposomal Formulations	206
Albumin-Bound Formulations	
Polymer-Bound Formulations	
Inorganic Nanoparticle Formulations	
Polymeric Nanoparticles	209
Dendrimer Formulations	210
Carbon Nanotubes Formulations	211
Gold-based Nanoparticle Formulation	212
NANOTHERAPEUTICS IN CNS DISORDERS	
Alzheimer's Disease	
Parkinson's Disease (PD)	
Multiple Sclerosis	
Amyotrophic Lateral Sclerosis (known as Lou Gehrig's Disease)	

Stroke	215
HIV-1-Associated Neurocognitive Disorders (HAND)	216
Prions Disease	216
NANOTHERAPEUTICS IN CARDIAC DISEASES	216
Imaging of Cardiovascular Diseases	217
Treatment of Cardiovascular Diseases	217
NANOTHERAPEUTICS IN INFECTIONS	218
NANOTHERAPEUTICS IN OCULAR DRUG DELIVERY	220
NANOTHERAPEUTICS IN RHEUMATOID ARTHRITIS	220
NANOTHERAPEUTICS IN THYROID	221
NANOTHERAPEUTICS IN VACCINES	223
SAFETY OF NANOTHERAPEUTICS	224
FUTURE PROSPECTS	228
CONCLUSION	220
CONSENT FOR PUBLICATION	22) 229
CONFLICT OF INTEREST	22) 229
ACKNOWLEDGEMENTS	22) 229
REFEREENCES	22)
	22)
CHAPTER 7 SCALE-UP, PRECLINICAL AND CLINICAL STATUS OF POLY (LACTIDE	2-
CO-GLYCOLIDE) AND ITS COPOLYMERS BASED DRUG DELIVERY SYSTEMS	246
Neha Bajwa, Preet Amol Singh, Kiran Jyoti, Ashish Baldi, Ramesh Chandra and	
Jitender Madan	
INTRODUCTION	247
CHEMISTRY AND PHYSICO-CHEMICAL PROPERTIES OF PLGA	247
PLGA Synthesis	247
Physicochemical Properties of Poly (Lactide-Co-Glycolide)	251
Solubility	251
Crystallinity	252
Thermal Stability	252
Mechanical Property	252
BIODEGRADATION, BIOCOMPATIBILITY, AND TOXICITY OF POLY (LACTIDE	-
CO-GLYCOLIDE)	252
Biodegradation Pathway	252
Biocompatibility	253
Toxicology	254
POLY (LACTIDE-CO-GLYCOLIDE) AND ITS COPOLYMERS BASED DRUG	
DELIVERY SYSTEMS	257
Microparticle Preparation Techniques	258
Method of Solvent Evaporation	258
Phase Separation Coacervation Method	258
Spray Drying	259
Nanoparticles Preparation Techniques	259
Implant Preparation Techniques	260
Solvent-Casting and Compression Molding	260
Extrusion	260
GelsConstruct Preparation Technique	260
Sponge Preparation Technique	261
Scaffolds Preparation Technique	261
PLGA Disk Preparation Technique	261
Electrocomposite Fiber Preparation Technique	262

DRUG RELEASE MECHANISMS FROM POLY (LACTIDE-CO- GLYCOLIDE) AND ITS COPOLYMERS BASED DRUG DELIVERY SYSTEM	
APPLICATIONS OF POLY (LACTIDE-CO-GLYCOLIDE) AND ITS COPOLYMERS IN	I
DRUG DELIVERY, VACCINE DELIVERY, AND MEDICAL DEVICES	
Vaccines and Immunology	•
Cancer	•
Cardiovascular Diseases	
Tissue Engineering	
Vascular Engineering	
Diagnosis	
Inflammation and Infection	
Contraceptives and Hormones	
INDUSTRIAL SCALE-UP OF POLY (LACTIDE-CO-GLYCOLIDE) AND ITS	
COPOLYMER BASED DRUG DELIVERY SYSTEMS	
PRECLINICAL STATUS OF POLY (LACTIDE-CO-GLYCOLIDE) AND ITS	
COPOLYMERS BASED DRUG DELIVERY SYSTEMS	
CLINICAL TRIALS OF POLY (LACTIDE-CO-GLYCOLIDE) AND ITS COPOLYMERS	3
BASED DRUG DELIVERY SYSTEMS	
US-FDA APPROVED POLY (LACTIDE-CO-GLYCOLIDE) AND ITS COPOLYMERS	
BASED DRUG DELIVERY SYSTEMS	•
CONCLUSION	
CONSENT FOR PUBLICATION	
CONFLICT OF INTEREST	
ACKNOWLEDGEMENTS	
REFERENCES	•
JECT INDEX	

## FOREWORD

Engineered nanomaterials (ENMs), due to their interesting physicochemical properties such as smaller size, larger surface area, electrical, optical and magnetic properties are being sought in a wide range of applications including technology, cosmetics, food packaging, medical imaging and drug delivery. Carbon nanotubes (CNTs), quantum dots, mesoporous and amorphous nanosilica, nanosilver, nano titanium and zinc oxides are some of the ENMs currently in commerce. Nevertheless, the attractive physicochemical characteristics of the ENMs also create concerns when exposed to, with respect to human and ecosystem health. This book on "nanomaterials" is very timely, and touches upon the different aspects of application of ENMs in drug delivery. The chapters in this book discuss the use of a spectrum of nanomaterials in drug delivery including nano metal oxides, CNTs and lipid nanoparticles, their various nanoforms, synthesis, characterization, efficacy in terms of drug delivery and the need for toxicity testing. Physicochemical characterization is an important aspect in nanotechnology, especially, in the realm of drug delivery. The synthesis of ENMs can introduce batch to batch variation in terms of size, shape, surface characteristics and chemical composition based on source materials and synthetic routes. Moreover, the stability of ENMs can be affected based on storage conditions. This book has thus given an importance to the aspect of physicochemical testing and discusses the different analytical methods to assess morphology, surface functionalities, behavior in solution, stability, etc. This book on "nanomaterials" also identifies the need for toxicity testing of the ENMs in drug delivery. Toxicity testing is a critical component for the selection of safer ENMs for application in drug delivery and to meet regulatory standards. This book has done a fantastic job in familiarizing the reader with the scope and application of the various ENMs and their nanoforms in drug delivery along with some insight into medical imaging and computational aspect of structureactivity relationships. I congratulate the editor Dr. Surendra Nimesh on doing a fantastic job with this book on the application ENMs in drug delivery, which is one of the promising emerging medical technologies.

#### Prem Kumarathasan

Analytical Biochemistry and Proteomics Laboratory Mechanistic Studies Division Environmental Health Science and Research Bureau Healthy Environments and Consumer Safety Branch Health Canada, Ottawa, Ontario, Canada K1A0K9

# PREFACE

Chemically synthesized drugs have been one of the major tools in combating several diseases, including bacterial and viral infections. However, these drug molecules face several barriers, including poor cellular uptake and instability in the physiological environment that masks the therapeutic potential. In order to circumvent these issues, there arises a need to develop vehicles that could effectively and safely transport the drug molecules to the target sites. Nanotechnology has come up as one of the potent and viable strategies. Several candidates have been proposed, such as nanoparticles, liposomes, carbon nanotubes, mesoporous silica nanoparticles, etc. These vectors can be modulated to achieve delivery, including drugs that are highly unstable and face difficulty in reaching sites. This book compilation brings together some of the eminent scientists working in different dimensions of nanotechnology. They have contributed chapters in their domain of knowledge that we believe would be highly useful not only for the young researchers but also for the experts looking for some exhaustive compilations.

Chapter 1 provides a detailed account of the application of lipid-based nanoparticles and nanostructures. This chapter also provides an overview of the recent literature on solid lipid nanoparticles and nanostructured lipid carriers for drug delivery applications. Background information on the origins, composition, characterization parameters, and biological applications of these nanocarrier systems has also been presented.

Chapter 2 provides an exhaustive account of the main route of preparation and applications of MSNs and silica nanomaterials. The chapter also provides insights into the chemistry, structure, and characterization of MSNs, followed by the synthetic strategies, and finally ends with a note on the application of MSNs.

Chapter 3 deals with hydrogels; they are defined as materials composed of water (hydro) and matrix (gel). The chapter discusses the role of polymer and peptide-based hydrogels, their multi-functionality, unique properties, and major uses. Hydrogels can serve as a major tool for human welfare in the future.

Chapter 4 talks about the application of metallic nanoparticles in drug delivery. Metallic nanoparticles have been used for treatment in some life-threatening diseases such as cancer. This chapter introduces gold and silver nanoparticles, nanoshells and nanocages, and their physicochemical properties. It illustrates some of the recent advances in the field of diagnostic imaging and cancer therapy.

Chapter 5 discusses the computational and experimental studies for the interaction of drugs with  $\beta$ -cyclodextrins. This chapter summarizes cyclodextrin's applications in drug delivery research through experimental and computational findings. In addition, it presents the highlights of various techniques of inclusion complex formations, mechanism of delivery systems, and their analytical methods.

Chapter 6 outlines the clinical applications of nanotechnology in various areas, including cancer, CNS disorder, rheumatoid arthritis, thyroid, cardiac diseases, ocular drug delivery, and vaccines. This chapter overviews the current status of pharmacological and clinical studies of nanoparticles in the development process.

Chapter 7 illustrates the scale-up, preclinical, and clinical status of PLGA, along with its copolymers-based drug delivery systems. This chapter summarizes the extensive applications, laboratory, and industrial-scale methods for the production of PLGA nano/microparticles, preclinical, and clinical status PLGA-based drug delivery systems.

#### Surendra Nimesh

Department of Biotechnology Central University of Rajasthan India

Nidhi Gupta Department of Biotechnology IIS (Deemed to be University) India

&

Ramesh Chandra Department of Chemistry University of Delhi India

#### iii

# **List of Contributors**

Ashish Baldi	Department of Pharmaceutical Sciences and Technology, Maharaja Ranjit Singh Punjab Technical University, Bathinda, Punjab, India
Charu Bharti	SD College of Pharmacy and Vocational Studies, Bhopal Road, Muzafrarnagar, UP Road, India
Chandrabose Selvaraj	Computer-aided Drug Designing and Molecular Modeling Lab, Department of Bioinformatics, Alagappa University, Karaikudi-630003, India
Délia Chaves Moreira dos Santos	Department of Pharmaceutics, Faculty of Pharmacy, Federal University of Minas Gerais, Belo Horizonte, Brazil
Faisal A. Almalki	Department of Pharmaceutical Chemistry, Faculty of Pharmacy, Umm Al- Qura University, Makkah Almukkarramah, Saudi Arabia
Gabriel Silva Borges	Department of Pharmaceutics, Faculty of Pharmacy, Federal University of Minas Gerais, BeloHorizonte, MG, Brazil
Guilherme Carneiro	Department of Pharmacy, Faculty of Biological and Health Sciences, Federal University of Jequitinhonha and Mucuri Valleys, Diamantina, MG, Brazil
Jitender Madan	Chandigarh College of Pharmacy, Mohali-140307, Punjab, India
Jyoti Dasharath Magare	C.S.M.S.S Dental College Kanchanwadi, Aurangabad, (M.S.), India
Kiran Jyoti	Department of Pharmaceutics, Chandigarh College of Pharmacy, Mohali (Punjab), India
Lucas Antônio Miranda Ferreira	Department of Pharmaceutics, Faculty of Pharmacy, Federal University of Minas Gerais, Belo Horizonte, Brazil
Mariana Silva Oliveira	Department of Pharmaceutics, Faculty of Pharmacy, Federal University of Minas Gerais, Belo Horizonte, Brazil
Muhammed Khalid Saifullah	Department of Pharmaceutical Chemistry, Faculty of Pharmacy, Umm Al- Qura University, Makkah Almukkarramah, Saudi Arabia
Monisha Singhal	Department of Biotechnology, IIS (Deemed to be University), Jaipur, Rajasthan, India
Neha Bajwa	Department of Pharmaceutical Sciences and Technology, Maharaja Ranjit Singh Punjab Technical University, Bathinda, Punjab, India
Nidhi Gupta	Department of Biotechnology, IIS (Deemed to be University), Jaipur, Rajasthan, India
Noushad Javed	Department of Pharmaceutics, School of Pharmaceutical Education and Research, Jamia Hamdard, New Delhi, India
Pradeep Kumar	Nucleic Acids Research Laboratory, CSIR-Institute of Genomics and Integrative Biology, Mall Road -110007, Delhi, India
Preet Amol Singh	Department of Pharmaceutical Sciences and Technology, Maharaja Ranjit Singh Punjab Technical University, Bathinda, Punjab, India
Ramesh Chandra	Dr. B.R. Ambedkar Centre for Biomedical Research, University of Delhi, Delhi, India Department of Chemistry, University of Delhi, Delhi, India

Ravi Goyal	Parexel International, Chandigarh, India
Rahul Goel	Nucleic Acids Research Laboratory, CSIR-Institute of Genomics and Integrative Biology, Mall Road -110007, Delhi, India
Reena Singh	Nucleic Acids Research Laboratory , CSIR-Institute of Genomics and Integrative Biology, Mall Road -110007, Delhi, India
Romila Manchanda	School of Basic and Applied Sciences, K.R. Mangalam University, Sohna Road, Gurugram, Haryana-122103, India
Rupinder K. Sodhi	Chandigarh College of Pharmacy, Mohali-140307, Punjab, India
Sabir Alam	NIMS Institute of Pharmacy, NIMS University, NH 11C, Delhi-Jaipur Expressway, Jaipur, India
Sanjeev Kumar Singh	Computer-aided Drug Designing and Molecular Modeling Lab, Department of Bioinformatics, Alagappa University, Karaikudi-630003, India
Seema Gupta	Department of Chemistry, Acharya Narendra Dev College, University of Delhi, New Delhi-110019, India

# Solid Lipid Nanoparticles and Nanostructured Lipid Carriers for Drug Delivery Applications

Gabriel Silva Borges<sup>1</sup>, Mariana Silva Oliveira<sup>1</sup>, Délia Chaves Moreira dos Santos<sup>1</sup>, Lucas Antônio Miranda Ferreira<sup>1</sup> and Guilherme Carneiro<sup>2,\*</sup>

<sup>1</sup> Department of Pharmaceutics, Faculty of Pharmacy, Federal University of Minas Gerais, Belo Horizonte, MG, Brazil

<sup>2</sup> Department of Pharmacy, Faculty of Biological and Health Sciences, Federal University of Jequitinhonha and Mucuri Valleys, Diamantina, MG, Brazil

Abstract: Lipid nanoparticles, such as solid lipid nanoparticles and nanostructured lipid carriers, are drug delivery systems in which solid lipids are dispersed in an aqueous phase stabilized by a surfactant layer. The great interest in these nanocarriers in the latest years is due to the biocompatible lipid matrix, associated with the potential for sustained drug release, and easy transposition to the industrial scale. Moreover, these lipid systems present the ability to prevent drug degradation, and to enhance cell uptake, usually increasing drug efficacy. This chapter will provide an overview of the recent literature on solid lipid nanoparticles and nanostructured lipid carriers for drug delivery applications. Thus, some background information on the origins, composition, characterization parameters and biological applications of these nanocarrier systems will be presented.

**Keywords:** Nanocarriers, Nanoparticles, Nanostructured lipid carriers, Solid lipid nanoparticles.

## **INTRODUCTION**

Nanotechnology is an exciting research field that, year after year, attracts more attention from researchers all over the world. It is defined as the research area that investigates nanometric systems, which are within the 1-1000 nm size range [1, 2].

In nanomedicine, nanoparticles are usually used for imaging, diagnosis and drug delivery purposes. Nanoparticles used for drug delivery are usually called nanocarriers. Nanocarriers can enhance the pharmacological activity, decrease

Surendra Nimesh, Nidhi Gupta and Ramesh Chandra (Eds.) All rights reserved-© 2021 Bentham Science Publishers

<sup>\*</sup> **Corresponding author Guilherme Carneiro:** Department of Pharmacy, Faculty of Biological and Health Sciences, Federal University of Jequitinhonha and Mucuri Valleys, Diamantina, MG, Brazil; E-mail: guilherme.carneiro@ufvjm.edu.br

#### 2 Nanomaterials: Evolution and Advancement (Part 2)

toxicity, and allow *in vivo* administration of drugs. There are many types of materials that can be used to produce nanoparticles for drug delivery, which include polymers, inorganic materials, and lipids [3]. In this context, this review focuses on novel lipid nanocarriers that have attracted much interest over the past 25 years: solid lipid nanoparticles (SLN) and nanostructured lipid carriers (NLC).

Both SLN and NLC are new generations of nanoemulsions (NE), being SLN the first generation (emerged in the 1990s) and NLC the second generation (emerged in the 2000s). Thus, we will discuss in this chapter the origin, key features, characterization and applications of these systems.

## BACKGROUND

Oil-in-water emulsions are conventional pharmaceutical dosage forms that are formed by oil droplets dispersed within an aqueous medium. Stabilization of oil droplets occurs by the use of surfactants that concentrate on the oil/water interface. Oil-in-water emulsions have been historically administrated by topical  $(e.g, \text{Diprolene}^{\$})$  and oral (e.g, mineral oil emulsion) routes [4].

Since the 1920s, scientists have examined forms of delivering emulsion by the intravenous route. The purpose of intravenous (IV) delivery of emulsions has been to provide energy and nutrients to hospitalized patients who cannot swallow foods normally. Emulsion droplet size can range from some nanometers to few micrometers, but this is not a limiting factor for the peroral and topical administration of emulsions. However, intravenous administration of particles with a size larger than a few micrometers can provoke vessel occlusion [5 - 7].

After years of research, in 1961, an IV fat emulsion (10% soybean oil stabilized with egg phospholipids) (Intralipid<sup>®</sup>) was released in Europe. The Intralipid<sup>®</sup> droplet sizes were around 276 nm. These small droplet sizes allowed Intralipid<sup>®</sup> to be delivered by the i.v. route. Since then, emulsions with narrow nanometric droplet size distribution (NE) started to be used for i.v. delivery of lipids [5, 8, 9].

In the beginning, NE were produced only to allow the delivery of oily components for hospitalized patients. A few years later, many drug-loaded NE arrived in the market (*e.g.*, Dizemuls<sup>®</sup>, Diprivan<sup>®</sup>, Etomidate-Lipuro<sup>®</sup>, among others) [10 - 12]. The success of these systems lies in the possibility of delivering hydrophobic compounds intravenously, but with no pain inconvenience [7, 13]. Moreover, NE present advantages such as toxicological safety and facile production on a large scale [14]. Drawbacks of NE systems, however, include low physical stability and low drug retention, leading to fast drug release and low drug stability. These drawbacks are due to the liquid nature of the lipids used in the NE [12, 15].

#### Nanoparticles and Nanostructured Lipid Carriers Nanomaterials: Evolution and Advancement (Part 2) 3

Liposomes represent another example of drug nanocarriers. Proposed in the 1960s by Bangham and co-workers [16], liposomes are, probably, the most well-known nanocarriers [17]. Phospholipid-based vesicles in the aqueous medium, the liposomes, entered the market in 1986 with Capture<sup>®</sup>, an anti-aging product by Dior<sup>®</sup> [18]. Later, the first pharmaceutical liposomes were approved: Alveofact<sup>®</sup> (1989), Ambisome<sup>®</sup> (1990), Doxil<sup>®</sup> (1995) and Daunoxome<sup>®</sup> (1996). These products explored the strategy of incorporating drugs into liposomal vesicles for some purposes, including better administration of poorly water-soluble drugs, enhancement of drug pharmacological effects and/or reduction of their toxicological effects [12, 18]. The major drawbacks related to liposomes are their low drug loading (DL) for hydrophobic drugs, difficulty in scaled-up production, and high production costs [10, 12, 15].

Polymers are another type of material widely used for nanocarriers production. The major drawbacks of polymeric nanocarriers include cytotoxicity, high cost of biodegradable polymers and scaled-up production difficulties [10, 12, 14, 15]. These drawbacks have hindered the insertion of polymer nanocarriers in the market.

Looking at all those aspects, there was a motivation towards the development of systems that could control drug release similarly to polymeric nanoparticles and liposomes, but without the cell toxicity, typically found in the former, and high production costs, as seen for the latter. In this context, SLN emerged as an alternative that could combine easy scaled-up production, fair costs, and biosafety of NE with the controlled release properties of liposomes and polymeric nanoparticles.

In the beginning of the 1990s, two researchers, Gasco and Müller, started working independently on the production of lipid nanocarriers that later would be called SLN. The first publication about SLN dates back to 1990 from Gasco's group [19]. It was followed by the first publication of Müller's group in 1992 [20]. Westesen's group has also worked on those lipid nanocarriers at that time [21].

Simultaneously with the papers, these researches yielded the first patents on SLN [22, 23]. While Müller's group used the term "solid lipid nanoparticles" since the beginning, Gasco's group used the term "lipospheres" [19, 24, 25] and later, "solid lipid nanospheres" [26 - 32]. Nowadays, the term "solid lipid nanoparticles" and its abbreviation, SLN, are consolidated.

SLN have the same constituents of a NE in a generic way (Fig. 1). They are formed by lipid droplets dispersed in an aqueous phase stabilized by a surfactant layer. The main difference is that, in the case of NE, the lipids that constitute the droplets are oils, *i.e.*, liquid lipids. In the case of SLN, the lipids used, mainly

## Silica Based Nanomaterial for Drug Delivery

# Charu Bharti<sup>1</sup>, Md. Sabir Alam<sup>2</sup>, Md. Noushad Javed<sup>3</sup>, Muhammed Khalid Saifullah<sup>4</sup>, Faisal A. Almalki<sup>4</sup> and Romila Manchanda<sup>5,\*</sup>

<sup>1</sup> SD College of Pharmacy and Vocational Studies, Bhopal Road, Muzafrarnagar, UP Road, India

<sup>2</sup> NIMS Institute of Pharmacy, NIMS University, NH 11C, Delhi-Jaipur Expressway, Jaipur, India

<sup>3</sup> Department of Pharmaceutics, School of Pharmaceutical Education and Research, Jamia Hamdard, New Delhi, India

<sup>4</sup> Department of Pharmaceutical Chemistry, Faculty of Pharmacy, Umm Al-Qura University, Makkah Almukkarramah, Saudi Arabia

<sup>5</sup> School of Basic and Applied Sciences, K.R. Mangalam University, Sohna Road, Gurugram, Haryana-122103, India

**Abstract:** The exceptional properties of mesoporous silica nanoparticles (MSNs) promote facile functionalization for improved drug delivery in nanotechnology. Recent advancements in this field have experienced the potential applications of MSNs and porous silicon (PSi). Tunable pore size, large surface area, and better surface properties make the materials possible to hold large amounts of payloads and prevent premature degradation. In this chapter, we will focus on and discuss the main route of preparation and applications of MSNs and silica nanomaterials.

**Keywords:** Drug delivery, Electron microscopy, Mesoporous silica nanoparticles (MSNs), Nanomaterials, Porous silica (PSi) nanoparticles.

#### **INTRODUCTION**

Nanoparticles are those particles whose dimensions range from 1-1000 nm, and small drugs and biological molecules, such as medicine, can be dissolved, entrapped, or bonded in a nanoparticle array. The bioavailability and distribution of nanoparticles can be easily modulated by changing the size range of nanoparticles [1, 2]. In the early 1990s, two independent research groups first invented mesoporous silica. The Mobile Oil Corporation first discovered MSNs in 1992, which were used in different fields due to their versatile textual properties like modifiable pore diameter and size distribution, large surface area, and volume

Surendra Nimesh, Nidhi Gupta and Ramesh Chandra (Eds.) All rights reserved-© 2021 Bentham Science Publishers

<sup>\*</sup> Corresponding author Romila Manchanda: School of Basic and Applied Sciences, K.R. Mangalam University, Sohna Road, Gurugram, Haryana-122103, India; E-mail:romila\_m@hotmail.com

of the pores. These attributes allow the encapsulation of a large number of active agents in the porous structure of MSNs [1, 3].

The attachment of various functional groups to the honeycomb-like porous MSNs can be facilitated and help to target drug molecules due to a large surface area as these MSNs are made up of many channels that are organized in a 2-D network. In 2001, Vallet- Regi *et al.* discovered MCM-41 and SBA-15 MSNs as delivery systems for hydrophilic and hydrophobic drugs, and later showed great interest in silica-based materials [4] due to their gate-keeping phenomenon and reactive polymers, which respond faster, and pH-sensitive materials were used to attach to the porous silica material [5]. Wang G and the group studied various processing parameters, such as pore size, connectivity, and geometry of MSNs, and checked how adsorption and release of ibuprofen could be affected [6]. The surface properties of MSNs, having silanol functionalization, could also control the drug loading and release profile of any agent.

MSNs exhibit superior biocompatibility at a suitable concentration for the pharmacological effect to other amorphous silica materials [7]. The highly permeable mesoporous structure of silica-based materials is used for payloads, whereas a hollow mesoporous structure has been widely used in a chemical reaction and controlled release [8].

## CHEMISTRY OF MESOPOROUS SILICA NANOPARTICLES

Due to high inertness, stable structure and hydrophilic nature silica-based systems show various advantages over organic polymers. These systems can therefore be easily dispersed in aqueous media and suitable for many water-soluble therapeutic agents [1, 3]. Silica material exhibits special characteristics such as it shows a good conjugation tendency to DNA or protein [9]. Due to the robust Si surface, emission at the Red light due to small size, amorphous silica nanoparticles also result in large oxidative stress cytotoxicity at 10-100µg/ml [10] and shows photostable and modifiable luminescence property [11].

The MSNs are mainly synthesized by chemically synthesized techniques from inorganic precursors such as tetra-alkoxysilanes and sodium silicates. Organic functional groups may also conjugate to the surface of MSNs by performing initial hydrolysis/condensation reactions using a tri-alkoxyorganosilane . Organo silanes such as 1,2-bis(tri-methoxysilyl)ethane and 1,4-bis(tri-ethoxysilyl)benzene react with each other in the synthesis medium of the core-shell of silicon dioxide which forms a surface on the shell after deposition of organosilica species [12].

#### Silica Based Nanomaterial

Chemically, an active honeycomb-like structure facilitates chemical functionalization, and modification of the surface leads to connect different therapeutic agents.

Various pore geometries of MSNs have been shown Fig. (1).



Fig. (1). Geometry of pores in mesoporous silica nanoparticles (a) Ia3d type (b) cubic pn3m type (c) cage type pm3n (d) cage type Im3 m.

MSNs show more protectiveness against external response *viz*. degradation and mechanical stress than liposomes, and dendrimers due to strong Si-bonds and it does not require any external stabilization for synthesis [18]. MSNs show different internal structures and orifice diameter represented in Table 1.

Table 1.	Different interior	• structures and	orifice diamet	er of MSNs.
I HOIC II	Different meetio	Sti actui co una	or mee anamee	<b>UI UI 11101 101</b>

Type of MSNs	Interior-Structure	Orifice-Diameter (nm)	References
MCM-50	Lamellar	2–5	[13, 14]
SBA-11	3D cubic	2.1-3.6	[14, 15]
SBA-12	3D hexagonal	3.1	[16, 17]
SBA-16	3D cubic structure - cage-like	4.0-9.0	[18]

## **CHAPTER 3**

## **Hydrogels for Drug Delivery**

Reena Singh<sup>1</sup>, Rahul Goel<sup>1</sup>, Seema Gupta<sup>2,\*</sup> and Pradeep Kumar<sup>1,\*</sup>

<sup>1</sup> Nucleic Acids Research Laboratory, CSIR-Institute of Genomics and Integrative Biology, Mall Road, Delhi-110007, India

<sup>2</sup> Department of Chemistry, Acharya Narendra Dev College, University of Delhi, New Delhi-110019, India

**Abstract:** The term 'Hydrogel' is self-defined, as a material composed of water (hydro) and matrix (gel). The hydrogels do not dissolve in water; rather, absorb water and swell into a volumetric mass due to their smart 3-dimensional network. Over the past few years, hydrogels have served as a multifunctional platform and gained the interest of the scientific community. The unique properties of hydrogels, including flexibility, biocompatibility, and mechanical stability, have made them quite an important research area in different fields like disease treatment, targeted drug delivery, and many others. The current applications of hydrogels include the manufacturing of contact lenses, drug delivery systems, hemostats, wound dressings, biosensors, *etc.* Here, the role of polymer and peptide-based hydrogels, their multi-functionality, unique properties, and major uses have been elaborated, which can serve as a major tool for human welfare in the future.

**Keywords:** Drug Delivery, Hydrogel, Polyethylene Glycol, Polymer, Nanocrystals.

## **INTRODUCTION**

In this era of developing novel biomaterials, rigorous research is being undertaken to contribute to various biomedical applications, including cell-mediated therapy, tissue regeneration, drug delivery, and so on. One group of these biomaterials that has become a rich area of interest is hydrogels. A highly hydrated 3D fibrous meshwork of peptides and polymers in hydrogels can be engineered to mimic the extracellular matrix (ECM) milieu – to support both *in vivo* and *in vitro* tissue engineering and provide *in vivo* drug delivery platform [1 - 5]. Hydrogel formation involves different types of interactions, *i.e.*, covalent, ionic, and weak

Surendra Nimesh, Nidhi Gupta and Ramesh Chandra (Eds.) All rights reserved-© 2021 Bentham Science Publishers

<sup>\*</sup> Corresponding author Pradeep Kumar and Seema Gupta: Nucleic Acids Research Laboratory, CSIR-Institute of Genomics and Integrative Biology, Mall Road, Delhi-110007, India and Department of Chemistry, Acharya Narendra Dev College, University of Delhi, New Delhi-110019, India; E-mails: pkumar@igib.res.in and seemagupta@andc.du.ac.in

#### Hydrogels for Drug Delivery

(including hydrophobic, van der wall forces, *etc.*) interactions. Hydrogels, being majorly composed of water (>90% water), are highly hydrophilic and play multifunctional roles in the hydrophilic physiological environment (Fig. 1).



Fig. (1). Representation of hydrogels, their assembly and applications.

Supramolecular self-assembly of peptides and polymers involves many noncovalent interactions that spontaneously and reversibly organizes the different peptides and polymers into functional and novel structures with the desired supramolecular nanoscale morphology. The use of this bottom-up approach for the synthesis of variable nanomaterials serves as a potential application in various fields including biomedical, different types of tissue engineering, and many other areas. The modification of functional groups over peptide building blocks and polymers using different chemical and biological methods followed by hydrogel formation creates an attractive platform for the scientific community to design and develop efficient and potential biomaterials.

Despite their use in different fields, hydrogels have encountered some limitations, viz., poor mechanical strength [6 - 8], low/non-uniform homogeneity, bio-

#### 92 Nanomaterials: Evolution and Advancement (Part 2)

#### Singh et al.

incompatibility, excessive volumetric swelling [9 - 12], low biodegradability / higher toxicity, expensive synthesis and fabrication [13, 14], which hinder its path to successfully function at the target site. To counter these problems, numerous hydrogels ranging from the long-chain peptide, *e.g.*, RADA16-I [15, 16] and RADKPS [17], to short peptide, *e.g.*, Leu- $\Delta$ Phe (Leucine  $\alpha,\beta$ - dihydroxy phenylalanine) [18], contributing as highly regulated drug delivery platforms for both hydrophilic and hydrophobic drugs have been reported in the past few years.

Hydrogels have been categorized based on their origin (natural and synthetic), assembly (spontaneous and stimulus-based), chain length (long [19] and short [20, 21]), and configuration (in case of peptides [22]) (Fig. 2). Both natural and synthetic polymers have been exploited to assemble into functional hydrogels. Natural polymers include chitosan, hyaluronic acid, sodium alginate, gellan gum, collagen, cellulose and gelatin, while the synthetic category includes polyethylene glycol (PEG), poly-lactic acid (PLA), polyvinyl alcohol (PVA), polyacrylic acid (PAA) and so forth [23, 24]. Broadly, the following sections deal with two types of hydrogels, *viz.*, polymer-based and peptide-based hydrogels.



Fig. (2). Types of peptides and polymers used for hydrogel formation under different stimulus conditions and their applications.

**CHAPTER 4** 

125

# Metallic Nanoparticles: Applications in Drug Delivery

Monisha Singhal<sup>1</sup>, Nidhi Gupta<sup>1</sup> and Jyoti Dasharath Magare<sup>2,\*</sup>

<sup>1</sup> Department of Biotechnology, IIS (Deemed to be University), Jaipur, Rajasthan, India <sup>2</sup> C.S.M.S.S Dental College Kanchanwadi, Aurangabad, (M.S.), India

Abstract: In the field of nanotechnology, the synthesis of metallic nanoparticles (NPs) is a demanding task for both modern phytopharmaceutical research as well as academics. These metallic nanostructures, NPs, made of metals such as gold and silver NPs can be used as quantum dots for the applications in Biomedical Science and Technology. Nanomedicine is a newly developed branch that is a boon for modern medicine. Nanotechnology in medicines will increase the production of the intended results and safety of the medicine. Nanoparticles are known for their stability, solubility, absorption and reduced toxicity. Metallic nanoparticles have been used for treatment in some life-threatening diseases such as cancer. This chapter introduces gold and silver nanoparticles, nanoshells and nanocages and their physico-chemical properties, and illustrates some of the recent advances in the field of diagnostic imaging and cancer therapy. Nanotechnology had a great influence on medical science and made a remarkable progress in the field of diagnostics.

**Keywords:** Cancer therapy, Diagnostic imaging, Metallic nanoparticles, Modern phytopharmaceutical, Nanotechnology.

## **INTRODUCTION**

Nanobiotechnology is the branch of biotechnology that deals with the synthesis and applications of several types of nanoparticles in biological systems. This technology has helped to design and develop better materials for medical applications in the areas of diagnosis as well as therapy. For generations, NPs have found their usage in routine applications such as in pottery and medicines. Nanotechnology has eventually made great progress in distribution and analysis in the field of biomedical science, as summarized in Fig. (1).

Surendra Nimesh, Nidhi Gupta and Ramesh Chandra (Eds.) All rights reserved-© 2021 Bentham Science Publishers

<sup>\*</sup> Corresponding author Jyoti Dasharath Magare: C.S.M.S.S Dental College Kanchanwadi, Aurangabad, (M.S.), India; E-mail: jyoti.magre@gmail.com

#### 126 Nanomaterials: Evolution and Advancement (Part 2)

Singhal et al.

Biocompatibility of metals can be enhanced with coatings of nanoparticles as nanofilms are known to have better interaction with living cells since nanoparticles films have a positive impact on cell adhesion and proliferation.



Fig. (1). Schematic diagram of various types of nanosystems in the field of drug delivery and diagnostics.

Nanoparticles are small in size, select a diseased site with the benefit of consuming only small amounts, thus producing fewer side effects. Nanoparticles in various forms and shapes such as micelles, vesicles and dendrimer capsules, have been utilized as drug delivery vehicles (Fig. 2). Nanotechnology has given the benefit of manipulating metals to the nanoscale range with remarkable properties and enhanced medical applications. Any structure designed and developed in the size range of 1 nm to 100 nm, in any one dimension is considered as nanoparticle [1].

## ADVANTAGES OF NANOPARTICLES-BASED DRUG DELIVERY

There are many advantages of using nanoparticles as drug carriers, as these increase the therapeutic efficiency and pharmacological features of drugs. For example, poor water-soluble drugs solubility could be improved by nanoparticles. The distribution of two or more drugs at once for the treatment of amorphous control and release of metabolic and therapeutic compounds might be enabled with the usage of nanoparticles [2, 3].

Nano-carriers for drug delivery system have several merits attached to them, such as:

1. Being very less volumetric, nanoparticles can easily move across the narrow capillaries.

#### Metallic Nanoparticles

- 2. Nanocarriers can penetrate the inter junction gap in between cells and tissues to reach the target organ.
- 3. The release of drugs can be controlled and therefore, drugs can be effective for a prolonged period.



Fig. (2). Various nanocarriers as drug delivery vehicles.

Owing to these features nano-based drug delivery vehicles have become favorable as carriers for the delivery of drugs than the conventional system [4]. The most important advantage of nanoparticles is in the preservation of pharmacological activity of the drug in which high doses of the drug can be incorporated without worrying about any modification due to chemical reactions [5].

## **APPLICATIONS**

- Achievement of targeted drug delivery therapy that can be used as a treatment for organs such as the brain and in diseased conditions like cancer;
- To deliver drugs and genes
- For detection of pathogens
- Proteins can be identified
- To map biomarkers
- Probing of DNA structure

## **CHAPTER 5**

# Computational and Experimental Binding Interactions of Drug and β-Cyclodextrin as a Drug-Delivery Vehicle

Chandrabose Selvaraj\* and Sanjeev Kumar Singh\*

Computer-aided Drug Designing and Molecular Modeling Lab, Department of Bioinformatics, Alagappa University, Karaikudi-630003, India

Abstract: cyclodextrins are primarily used to enhance the aqueous solubility and stability of drug molecules and they can be chemically modified to display functional groups on their primary or secondary rim. It belongs to the cyclic polymers ( $\alpha$ -1,4) linked oligosaccharides of  $\alpha$ -glucopyranose units with hydrophobic inner core and hydrophilic outer surface. This combination of functionality and guest binding ability makes Cyclodextrin is an important scaffold to design functional supramolecular systems. Due to its structural characteristics, it can interact with appropriately sized drug molecules to form an inclusion complex. Inclusion into the Cyclodextrin's cavity alters the physicochemical properties of an included compound, especially on increasing its dissolution rate and sometimes in increasing the drug inhibition rate. Structural factors raise favor for this non-covalent inclusion complex and offer a variety of pharmaceutical applications that may be used in many industrial products. The negligible cytotoxic effects of cyclodextrin are an important attribute in applications such as drug carriers, food and flavors, cosmetics, packing, textiles, separation processes, environment protection, fermentation, and catalysis. Through this chapter, we aimed to summarize Cyclodextrin's applications in drug delivery research through experimental and computational findings. In addition, we tried to present the highlights of various techniques of inclusion complex formations, mechanism of delivery systems and their analytical methods.

**Keywords:** Drug Delivery, Drug, Drug-Polymer Interactions, Drug Carriers, β-Cyclodextrin, Inclusion Complex, SEM, TEM.

#### INTRODUCTION

The classical natural-based drug development is an incredibly tedious process and so the current revolution in drug discovery is driven by functional genomics, pro-

Surendra Nimesh, Nidhi Gupta and Ramesh Chandra (Eds.) All rights reserved-© 2021 Bentham Science Publishers

<sup>\*</sup> Corresponding authors Chandrabose Selvaraj and Sanjeev Kumar Singh: Computer-aided Drug Designing and Molecular Modeling Lab, Department of Bioinformatics, Alagappa University, Karaikudi-630003, India; E-mails: selnikraj@bioclues.org and skysanjeev@gmail.com

#### 152 Nanomaterials: Evolution and Advancement (Part 2)

teomics, high throughput screening and *in silico* drug designing. Recently, several strategies have attempted to improve drug safety and efficacy, such as individualization of drug therapy, dosing, monitoring and controlled administration [1, 2]. The selective distribution of drug molecules in their position should increase their efficacy while minimizing adverse side effects. Referable to the solubility, distribution and duration of action, the delivery of some drug molecules becomes challenging and the continuous actions may be modified by a combination of drug vehicle molecules. Drug delivery vehicles are also designated as drug carriers that act as the most important entity required for the successful passage of loaded drug molecules. The ideal drug delivery vehicle would be targeting all individual cells and should possess unique characteristic features such as biocompatible and virtually 100% reliable, able to cross bloodbrain barriers and also known by the target cells and must preserve the specificity surface ligands [4, 5].

Currently, various carrier molecules such as liposomes, monoclonal antibodies, fragments of modified plasma proteins, microspheres, lipoproteins and nanoparticles are constantly intended to overcome the intelligible properties of drug particles. Liposomes are small artificially designed vesicles composed of bilipid membrane specially designed to bind and rapidly taken by macrophages for passive drug delivery (Fig. 1); they have been extensively used in non-viral vector-mediated gene therapy [5]. Monoclonal antibodies are a type of glycoprotein that include polypeptide and carbohydrate according to the ratio of 82:18 and 96:4, respectively. The drug complex with monoclonal antibody provides selective targeting for tumor cell masses or lymphomas. Modified plasma proteins are a unique method of transporting drugs due to their solubility and relatively small molecular weight. It can be easily modified by attaching different molecules such as peptides, sugars and other ligands to an appropriate mode of drug delivery. Microspheres are soluble polymers ideally having a particle size less than 200 µm and well planned to subdue some of the troubles of conventional therapy and enhance the healing efficacy of drugs. Microspheres are classified into two characters, namely synthetic polymers and natural polymers. Various techniques for the carrier particles that include solubilization solid dispersion are effectively applied to enhance the bioavailability and drug solubility [6 - 8]. Even though these methods show reliable applications and have some disadvantages such as low drug loading and heavy doses. Moreover, the complexation of drug-polymer interactions came up with the Drug-cyclodextrin inclusion complex as a novel and alternate method for drug delivery system [9, 10]. Cyclodextrins are cyclic oligosaccharides having lipophilic inner cavities that can trap or encapsulate other molecules and hydrophilic outer surfaces capable of interacting with a big assortment of molecules to make non-covalent inclusion complexes to lead a host-client type of complexity that can change or improve the β-Cyclodextrin as a Drug-Delivery Vehicle

physical, chemical and biological properties of guest molecules.



Fig. (1). Cartoon representation of typical polymeric micelles and liposomes.

## **Chronological Evolution of Cyclodextrin**

The discovery of cyclodextrin was investigated in the late 19th century by Antonie Villiers during his research on the action of enzymes on various carbohydrates, particularly using the butyric Bacillus amylobacter on potato starch. Through this research, he revealed that under suitable conditions with incubation treatment, potato starch could ferment and produce dextrin because of the action of Bacillus microbes. In summary, he also discovered other byproducts that can be obtained after a few days of incubation, a highly crystalline substance with a composition of starch and dextrin. When examining the properties of dextrin, there are many similarities to cellulose, called cellulosin, which is resistant to acid hydrolysis and the absence of reducing sugars [11, 12]. Subsequently, manipulation of experimental conditions resulted in the two new distinct crystallised cellulosins, alpha-dextrin (a-cyclodextrin) and beta-dextrin (β-cyclodextrin). From 1853 to 1920, Franz Schardinger studied heat-resistant microorganism that can contribute to food poisoning and he found a type of extremely heat resistant microorganism that can dissolve starch from crystalline byproducts remarkably like the cellulosine reported by Villiers [13, 14]. He was the first person who explained the underlying properties of dextrins and later he proposed that dextrins were cyclic polysaccharides. Freudenberg confirmed long predictions of dextrin in the form of cyclic polysaccharides after 30 years, because these were cyclic oligosaccharides [15]. In 1911, Hans Pringsheim, a German chemist, played a critical role in research on dextrins and repeated the same experiments of Schardinger and isolated pure alpha dextrin and beta dextrins and

# **Clinical Milestones in Nanotherapeutics: Current Status and Future Prospects**

Ravi Goyal<sup>1</sup>, Rupinder K. Sodhi<sup>2,\*</sup> and Jitender Madan<sup>2</sup>

<sup>1</sup> Parexel International, Chandigarh, India

<sup>2</sup> Chandigarh College of Pharmacy, Mohali-140307, Punjab, India

**Abstract:** An aging population and poor clinical solutions for several diseases have propelled the rapid emergence of nanotherapeutics. Advanced drug delivery has turned out to be an important aspect of the medical field. A targeted delivery system transports the drug to the place of action hence, minimizing its adverse side effects on other vital tissues. Cell-specific targeting can be achieved by coupling drugs to specially framed carriers. Various nanoparticles, including solid lipid nanoparticles, nanosuspensions, nanoliposomes, micelles, polymeric nanoparticles, magnetic nanoparticles, dendrimers, carbon nanotubes, and fullerenes have been developed as carriers in drug delivery systems. In this chapter, the aforementioned nanocarriers and their clinical milestones achieved in various arenas including cancer, CNS disorder, rheumatoid arthritis, thyroid, cardiac diseases, ocular drug delivery, and vaccines so far, are scrutinized. This chapter outlines the current status of pharmacological and clinical studies of nanoparticles in the development process.

**Keywords:** Alzheimer's disease, Cancer, Carbon nanotubes, Central nervous system, Clinical trials, Drug delivery, Fullerenes, Magnetic nanoparticles, Nanosuspensions, Nanoparticles, Nanoliposomes, Parkinson's disease, Vaccine.

## **INTRODUCTION**

Nanotherapeutics is one of the most advanced controlled drug delivery systems (CDDS) [1]. Nanoformulations are definitely among the promising carriers for the new era of drug delivery systems where therapeutic agents can be delivered selectively to target sites. Nanoformulations offer the advantages of being longer acting, rapidly soluble, less toxic, less immunogenic, and better than controlled-release formulations [2]. Targeted delivery in nanotechnology can be achieved by two approaches bottom-up and top-down fabrication at a nano-level with size and functional group modifications [3]. The therapeutic combat the disease at a

Surendra Nimesh, Nidhi Gupta and Ramesh Chandra (Eds.) All rights reserved-© 2021 Bentham Science Publishers

<sup>\*</sup> **Corresponding author Rupinder Kaur Sodhi:** Department of Pharmacology, Chandigarh College of Pharmacy, Mohali; Punjab; India; Email: ccp.rupinder@cgc.edu.in

#### **Clinical Milestones**

minimum required dosage without causing significant adverse events, thereby improving the quality of life (QoL) of the patient. All these objectives could be met with the application of nanotechnology to therapeutic agents. Stability and safety are the major concerns for these nanoformulations. The blood-brain barrier (BBB) is the most critical barrier in designing drugs for the central nervous system (CNS) targeting. With the advancement in nanotechnology, newer techniques have been developed to stabilize these molecules and also make them more selective hence, reducing their toxicity [1, 2]. Trials dedicated to the toxicology of these agents should be carried out as a part of preclinical studies as well as clinical studies to establish the safety of these formulations.

## **DRUG TARGETING**

The optimization of the nanotherapeutic agent depends on the conjugation of the drug to the carrier. A drug molecule could be covalently attached to the carrier, adsorbed on to the carrier, or encapsulated into the carrier. For the cell-specific targeting, covalent linking provides advantages over other strategies as the amount of drug (number of drug molecules) attached can be controlled. Either passive or active mechanisms could be used for cell-specific drug targeting. Passive drug targeting can be achieved through enhanced permeability and retention (EPR). Effective drug targeting is a function of the altered physical stimulus (viz. pH, temperature, and magnetism) [1, 4]. It can be achieved by targeting cell-specific ligands and receptors. At the target sites, drugs are released from the carriers and the release of the drugs could be controlled by altering the physiological parameters, *e.g.*, pH, osmolarity, temperature, and enzymatic activity [4]. The ultimate aim of the targeted drug delivery is to ensure prolonged, localized, and protected drug action within impaired tissue.

### **CARRIERS FOR NANOTHERAPEUTIC FORMULATIONS**

Nanotherapeutic formulations are better than conventional delivery systems due to enhanced site-specific delivery owing to the size of nanoparticles (NPs) [4]. Cellspecific drug targeting can be achieved by a combination of specially designed carriers and drug molecules [1]. The uptake of nanocarriers by cells is dependent on the physicochemical and biological properties of these carriers. Optimized nanocarriers have better penetration into the cell and have a targeted delivery compared to the larger molecules. A wide range of advanced nanocarriers have been used in therapeutics viz. solid lipids nanoparticles, polymeric nanoparticles, liposomes, micelles, dendrimers, nanosuspensions, carbon nanotubes, fullerenes, and magnetic nanoparticles [1, 4]. The scope of clinical applications of these nanoparticles has been described in detail.

## **Solid Lipids Nanoparticles**

Solid lipid nanoparticles (SLNs) have been characterized as an alternative drug delivery system to other colloidal carriers, such as emulsions, liposomes and polymeric micro- and nanoparticles. Lipids mainly comprise triglycerides (tricaprin, trilaurin, and tripalmitin), partial glycerides (Imwitor), fatty acids (palmitic acid, stearic acid), steroids (cholesterol), and waxes (cetyl palmitate). Numerous emulsifiers could be used to stabilize the aqueous lipid dispersion according to their molecular weight and charge. The quality and quantity of the surfactant elected to depend upon the selected lipid and route of administration. Commonly used surfactants include egg lecithin, soybean lecithin, sodium cholate, and Poloxamer 188 [5]. From a regulatory perspective, the delivery system used should be chemically and physically stable. SLNs offer several advantages of being easy to prepare; chemically and physically stable, and control release properties due to the EPR effect [6]. They can carry a variety of molecules to the target location including proteins, antigens, nucleotide, and small drug molecules [7]. SLNs have shown significant efficacy for drug targeting to the brain as these molecules can easily cross BBB [8]. SLNs are also potential vehicles for gene therapy due to their chemical stability. SLNs can be used for transdermal drug delivery to selective skin layers due to its drug modulation properties [9]. Nanostructured lipid carriers (NLC) are advanced SLNs that have better physicochemical properties than conventional SLNs. A large number of SLNs with a variety of anticancer drugs (doxorubicin (DOX), camptothecin (CPT), etoposide, paclitaxel (PTX), idarubicin, retinoic acid, and SN-38 [irinotecan analog]) have been prepared and tested in animals; however, clinical trial of these formulations has to be carefully designed to establish safety and efficacy in humans [6]. Manjunath and Venkateswarlu conducted a study in which tripalmitin SLNs were loaded with clozapine and after intravenous (IV) administration in mice, there was a significant increase in drug concentration in the brain compared to clozapine suspension. Similarly, after intraperitoneal administration of tripalmitin SLNs of etoposide, increased brain drug levels were observed compared to conventional etoposide [10]. In a study by Yang and coworkers, Stearic acid SLNs stabilized with Poloxamer 188 and loaded with CPT showed targeted drug delivery to the brain by crossing BBB both via intravenous and oral routes. Maximum serum concentration (C<sub>max</sub>), Area under the curve (AUC)/dose and mean retention time were higher by 180%, 10.4 folds and 4 folds, respectively than the drug solution [11]. Singh and coworkers achieved increased drug targeting of docetaxel to brain tumors with excellent safety and efficacy by conjugation of Lactoferrin (*Lf*) on SLNs surface [12]. SLNs could be used for drug delivery to the lungs. Dry powder formulations and aerosols for inhalation could be prepared using spray drying with lactose as excipient [13, 14].

## **CHAPTER 7**

# Scale-up, Preclinical and Clinical Status of Poly (Lactide-Co-Glycolide) and its Copolymers based Drug Delivery Systems

Neha Bajwa<sup>1</sup>, Preet Amol Singh<sup>1</sup>, Kiran Jyoti<sup>2</sup>, Ashish Baldi<sup>1</sup>, Ramesh Chandra<sup>3,4</sup> and Jitender Madan<sup>2,\*</sup>

<sup>1</sup> Department of Pharmaceutical Sciences and Technology, Maharaja Ranjit Singh Punjab Technical University, Bathinda, Punjab, India

<sup>2</sup> Department of Pharmaceutics, Chandigarh College of Pharmacy, Mohali (Punjab), India

<sup>3</sup> Dr. B.R. Ambedkar Centre for Biomedical Research, University of Delhi, Delhi, India

<sup>4</sup> Department of Chemistry, University of Delhi, Delhi, India

Abstract: Poly(lactide-co-glycolide) or PLGA is a kind of a synthetic polymer that has been approved by USFDA for its use in humans. PLGA nano/microparticles have proved to offer controlled as well as the sustained release of several medicinal moieties. PLGA is chemically synthesized by direct polycondensation of glycolic acid (GA) with lactic acid (LA) and different factors like LA: GA ratio, storage temperature, the initial molecular weight of the monomers, and exposure time to water influences the physical properties of PLGA. Similarly, various factors like morphology, crystallinity, molecular weight, shape, size molecular, hydrophobicity, chemical structure, physicochemical properties, glass transition temperature are some of the crucial factors responsible for the biodegradation of PLGA. PLGA based micro/nanoparticles are generally prepared by the oil-in-water emulsification process. On the other hand, spray drying is one of the industrial methods for the production of PLGA particles. In this chapter, we have summarized the extensive applications, laboratory, and industrial-scale methods for the production of PLGA nano/microparticles, preclinical, and clinical status PLGA based drug delivery systems.

**Keywords:** Industrial scale-up, Microparticles, Nanoparticles, Physicochemical properties, Poly(lactide-co-glycolide), Preclinical and clinical status, Scaffolds, Vaccine adjuvant.

Surendra Nimesh, Nidhi Gupta and Ramesh Chandra (Eds.) All rights reserved-© 2021 Bentham Science Publishers

<sup>\*</sup> Corresponding author Jitender Madan: Department of Pharmaceutics, Chandigarh College of Pharmacy, Mohali-140307, Punjab, India; E-mail: jitenderpharmacy@gmail.com

## **INTRODUCTION**

Recently, extensive consideration has been given to the utilization of polymers in drug delivery. Polymers such as polycaprolactone, alginate, and chitosan have been discovered and widely utilized for drug delivery. Polymers are obtained both from natural and synthetic origins and possess different merits and demerits. Merits of natural polymers include good cell adhesion and cell function in comparison to synthetic polymers while limited purity and allergenic reactions are the demerits of natural polymers. In contrast, among natural [1 - 6] and synthetic polymers, synthetic polymers viz. poly (amide), poly (amino acid), and poly (ester) are extensively used in the customization of drug delivery carriers [7, 8]. Poly (lactide-co-glycolide) (PLGA), is a kind of synthetic polymer that has been extensively explored to achieve desired effects or targets in drug delivery. Additionally, PLGA possesses the unique merit of being a biodegradable polymer, which is synthesized through direct polycondensation of glycolic acid (GA) with lactic acid (LA). Random opening of ring and copolymerization of monomers like cyclic dimer (1,4-dioxane-2,5-diones) of lactic acid and glycolic acid, in the company of catalyst especially tin (II) 2-ethyl hexanoate, tin (II) alkoxide, or aluminum isopropoxide results in the formation of PLGA [9, 10].

Foremost work was carried out on PLGA in the area of biomedicine from 1960 to 1970 period. Subsequently, the investigation of polymer led to its extensive use in several fields, *e.g* vascular graft, dental and fracture repair, gene delivery, and cancer therapy [11, 12]. Furthermore, USFDA approved PLGA for its use in various drug delivery systems *i.e.* microspheres, microparticles, implants, nanoparticles, and *in-situ* gel formation. PLGA microparticles are used for tendering controlled and sustained release of Human Growth Hormone (HGH) and for the treatment of cancer, cardiovascular diseases, infection, and inflammation [13, 14].

## CHEMISTRY AND PHYSICO-CHEMICAL PROPERTIES OF PLGA

## PLGA Synthesis

Poly (lactide-co-glycolide) is synthesized by the method of direct polycondensation of glycolic and lactic acid (Scheme 1). The polycondensation reactions are responsible for the solution and the melted or solid-state of the polymer. The manufacturing of high molecular weight polymer requires a high degree of dehydration which is difficult to accomplish and hence, polycondensation is considered as an incomplete method to get high molecular weight polymers [15]. However, synthesis of high molecular weight (160 kDa) PLGA was achieved by azeotropic dehydration glycolic acid and L-lactic acid

#### 248 Nanomaterials: Evolution and Advancement (Part 2)

Bajwa et al.

solution in diphenyl ether at 130°C in presence of tin powder as a catalyst for 20-40 h [16]. For the complexity and purification of the end product, the solvents were introduced; hence, polycondensation is performed to obtain the melting state of the polymer. The melt polycondensation method possesses two equilibria: the ring/chain equilibrium for the formation of cvclic diester and hydration/dehydration for the formation of ester. However, in typical melt polycondensation, glycolide, and cyclic diesterslactide (Scheme 1) are synthesized by the depolymerization method. The specific condition of this reaction *i.e.* high temperature under vacuum, can not only produce dehydration but also forms cyclic diesters in equilibrium with the polymer. This reaction repulses the excessive growth of the chain of the polymer, and thus it cannot be employed for high molecular weight polymer. To surmount this problem, the method involving melt/solid polycondensation was recently developed to achieve a specific objective (Scheme 1) [17].



**Scheme 1.** Steps involved in the synthesis of poly (lactide-co-glycolide, PLGA).

In the first step, dehydration of monomers can formulate the oligomers, followed by the second step, where oligomers are under vacuum (10 Torr), and tin chloride dehydrate/p-toluenesulfonic acid or zinc acetate dihydrate can be used as a

## **SUBJECT INDEX**

## A

Absorption spectroscopy 138 Acid 10, 11, 14, 19, 22, 31, 37, 38, 60, 61, 68, 79, 92, 96, 97 103, 105, 114, 115, 131, 157, 169, 178, 196, 198 201, 204, 209, 210, 213, 219, 221, 222 246, 247, 251, 252, 261, 269 aspartic 114 behenic 10 butyric 14 citric 131. 169 deoxyribonucleic 201 docosahexaenoic 22 docosanoic 10 flufenamic 178 folic (FA) 31, 37, 38, 131, 157 glutamic 105 glycolic (GA) 19, 68, 210, 246, 247, 251, 252, 261, 269 hyaluronic 37, 92, 96, 204, 210 lactic 198, 209, 219, 246, 247 lactobionic 61 Octadecanoic 10 oleic 11, 22 poly-gamma glutamic 115 polvlactic 103 retinoic (RA) 22, 196, 221, 222 salvianolic 79 sialic 213 Sulfonic 60, 61 synthesized thiolated hyaluronic 97 tricarboxylic 269 Activity 31, 32, 36, 37, 93, 94, 95, 100, 101, 113, 117, 134, 139, 142, 197, 203, 204, 211, 215, 218, 219, 256, 267, 272 antibacterial 93, 272 anti-cancerous 134, 267 antifungal 134 anti-HIV 197 anti-metastatic 142 antimicrobial 100

antioxidant 204 antiviral 219 bacteriostatic 256 cell-mediated immunity 117 effective fungicide 139 electron transfer 113 hepatoprotective 203 Adenocarcinoma 201, 207, 227 inoperable metastatic 201 metastatic pancreatic 207 AIDS-related Kaposi's sarcoma 227 Albumin 104, 129, 165, 198, 208 bovine serum 104 human serum 165, 208 Alginate 92, 98, 113, 130, 198, 247 -fibrin microfibers 98 natural polysaccharide 130 Allergic 198, 214 encephalomyelitis 214 rhinitis 198 Alpha-tocopheryl succinate 22 Alport syndrome 181 Alzheimer's disease 140, 181, 194, 199, 212 Amyotrophic lateral sclerosis 215 Analysis 27, 94, 105, 106, 110, 175 gene expression 94, 106, 110 microscopic 27 rheological data 105, 106 thermal evolution 175 Angiogenesis 106, 218, 222, 269 synovial cell 222 Antibacterial 134, 137, 219, 221 agents 133, 143 drug therapy 218 efficacy 218, 219, 220 hydrogel for wound Healing 95 Anticancer 141, 162, 167, 221, 222 activity 141, 167, 221, 222 therapy 162 Antigen presenting cells 265 Anti-inflammatory activities 140, 222 Antilipidemic effect 31

Surendra Nimesh, Nidhi Gupta and Ramesh Chandra (Eds.) All rights reserved-© 2021 Bentham Science Publishers

#### 294 Nanomaterials: Evolution and Advancement (Part 1)

Apoptosis 142, 143, 203, 206 inhibiting 203 Apoptotic death 141 Application 132, 138, 202 of AgNPs 138 of AuNPs 132 of gold nanoparticles 132 of nanosuspensions 202 Arterial occlusive disease 268, 276 Artificial skin transplantation 93 Aspergillus 134 fumigatus 134 niger 134 Atherosclerosis 181 Atomic force microscopy (AFM) 27, 28, 62, 99.109 Atom transfer radical polymerization (ATRP) 97 ATP binding cassette (ABC) 206 Attenuated total reflectance (ATR) 106

#### B

Bacillus 131, 180, 219 circulans 180 macerans 180 subtilis 131, 219 Bacterial enzyme cyclodextrin glycosyltransferase 179 **Biodegradation pathway 252** Biomaterials 69, 90, 93, 94, 110, 255, 264, 267, 268 engineering scaffold 110 interconnected scaffold 94 Biosensors 71, 77, 78, 90, 92, 113, 117, 128, 138, 140, 141, 204 electrochemical 113 herbicides detection 138 Bone marrow stem cells (BMSCs) 93, 98 Bovine serum albumin (BSA) 104, 130 Bradykinesia 213 Brain 38, 39, 196, 197, 203, 209, 212, 213, 215, 216, 275 cell death 203

ischemic 197 Breast cancer 37, 76, 167, 205, 207, 208, 211, 227 and diffuse large B-cell 227 diagnosing 76 Breast carcinoma 207, 210, 227 Buserelin apogepha prostate cancer 277

## С

Calcium phosphate cement (CPC) 98 Camellia sinensis 138 Cancer 36, 71, 72, 76, 78, 125, 133, 136, 166, 167, 168, 194, 201, 203, 204, 205, 206, 207, 208, 209, 210, 212, 226, 227, 247, 265, 266, 267 advanced urothelial 209, 226 binary 133 cardiac 168 cervical 168, 210 epithelial 76 fallopian tube 212 gastric 210, 227 gynecologic 210 hematopoietic 210 hepatocellular 207 imaging brain 206 immunotherapeutics 265 non-muscle-invasive bladder 208 stomach 205 therapy 71, 72, 125, 136, 247 xenografts 209 Cancer treatment 16, 74, 101, 141, 166, 201, 247, 266 strategy 141 Candida albicans 131, 219 Carcinogenesis 207 Carcinomas 205, 206, 207, 208, 209, 210, 211.227 adrenocortical 207, 227 esophageal 205 ovarian 209, 227 pancreatic 208, 227

urothelial 210

Nimesh et al.

#### Subject Index

Cardiovascular diseases 216, 217, 247, 267 Catalysis 136, 142, 151 Central nervous system (CNS) 38, 39, 194, 195, 198, 275 disorder 275 Central venous catheter (CVC) 139 Cerebral 38, 215 ischemia 215 parenchyma 38 Chemical vapour deposition 68 Chemistry of cyclodextrin 162 Chemotherapy regimens 207 Chitosan 30, 39, 92, 93, 198, 199, 219, 220, 247 Cholangiocarcinoma 210 Cholera toxin 101 Chondrogenesis 204 Chronic 11, 107, 167, 212, 228, 275 immunity 107 leukemia 167 lymphocytic leukemia (CLL) 212, 228 periodontitis 275 toxicity 11 Circular dichroism (CDs) 103, 110, 161, 164, 167.180 Combustion synthesis process 67 Compression molding technique 261 Computed tomography (CT) 76, 217 Contact lenses 90, 104 nanoparticle-loaded hydrogel-based 104 Controlled drug delivery systems (CDDS) 117, 194 Copolymers based drug delivery systems 257, 271, 275, 276 Co-precipitation 157 method 157 techniques 157 Corneal retention time 220 Coronary 276 artery lesions 276 disease 276 Coupled plasma mass spectrometry 70 Critical flocculation temperature (CFT) 62 Crystalline mesoporous nanostructure 66 Cyclodextrin-based 161, 166

Nanomaterials: Evolution and Advancement (Part 1) 295

supramolecular nanotechnology 166 supra-molecular vesicles 161 Cyclodextrin system 165 Cytological aspiration 257 Cytotoxicity 3, 37, 58, 76, 142, 224 large oxidative stress 58

#### D

Degradation 4, 11, 15, 21, 34, 35, 94, 96, 102, 114, 163, 166, 170, 172, 208, 249, 251, 252, 253, 254, 255 acidic 255 biological 15 chemical 11 collagen 94 enzymatic 34, 35, 163, 208 metabolic 34 oxidative 170 protease 114 thermal 249, 252 Degradation kinetics 253 Dementia 198, 199 Demerits 247 Demyelinating 214 Dendrimer-enhanced oxaliplatin (DEO) 210 Dendrimeric polyguanidilyated translocators (DPTs) 220 Dendritic cells (DCs) 224, 264, 265 Deposition 58, 93 electrophoretic 93 Devices 17, 28, 72, 113, 133, 136, 138, 139, 259, 263, 268, 275, 276 medical 136, 263 memory 133 microfluidic 259 prosthetic 268 ultrafiltration 28 ultrasonication 17 Diabetes mellitus 133 Differential scanning calorimetry (DSC) 26, 63, 175 Diffuse large B-cell lymphoma (DLBCL) 207, 227

#### 296 Nanomaterials: Evolution and Advancement (Part 1)

Dip coating technique 66 Diseases 72, 76, 78, 103, 106, 107, 125, 129, 133, 139, 141, 181, 194, 198, 203, 214, 215, 216, 220, 221, 223, 224, 267 269 autoimmune 221 bowel 269 cardiac 194, 216 chronic inflammatory 269 congenital neuro-inflammatory 220 fatal 267 genetic 72 inflammatory 214 life-threatening 125 neurodegenerative 198, 203, 214, 215 nonalcoholic fatty liver 181 Disorders 204, 205, 212, 214, 216 chronic kidney 205 dermal 204 neurological 214 Disseminated intravascular coagulation (DIC) 225 DNA 38, 140, 210 replication in cancer cells 210 /RNA detection 140 synthesis 38 Docking methods 165 Drug delivery 181, 196, 197, 202, 271, 274, 276 anti-inflammatory 274 mucosal 271 oral 181 polymeric 276 pulmonary 202 transdermal 196, 197 Drugs 72, 73, 114, 132, 133, 134, 135, 143, 152, 197, 198, 199, 211, 216, 219, 267 anti-cancer 72, 73, 135, 143, 267 antidiarrheal 199 anti-inflammatory 114, 197 anti-leukemic 134 anti-microtubule 211 anti-parasitic 219 antiretroviral 199, 216 antirheumatic 133 antiviral 199

immuno-modulator 199 pulmonary 198 therapeutic 132 topoisomerase inhibitor 211 transporting 152 Dynamic light scattering (DLS) 24, 25, 110

#### Е

Edema 215 Effect 3, 96, 143, 203, 254 carcinogenic 254 cytotoxic 143 drug pharmacological 3 hypoglycemic 96 synergistic immunomodulatory 203 Efficacy 39, 152, 165, 166, 196, 203, 204, 205, 207, 208, 210, 214, 216, 217, 220, 222 bactericidal 220 Efficiency 28, 34, 100 encapsulated 100 encapsulation process 28 inflammatory 34 Efficient 36, 100, 110, 112, 116 anticancer treatment 36 cytocompatibility 100, 110, 112, 116 Electrohydrodynamic spraying 100 Electrophoretic light scattering (ELS) 25 ELISA 102, 133 analysis 102 tests 133 Emission 58, 76, 217 single-photon 76, 217 Emulsification method 271 Emulsifiers 8, 12, 14, 15, 196, 258, 271 low molecular weight 271 rational choice 15 Emulsions 2, 8, 16, 20, 35, 66, 131, 196, 258, 260, 262, 271 injectable 35 mineral oil 2 prepared nanoparticle 131 water-in-oil 260

#### Nimesh et al.

#### Subject Index

Encapsulated resperidone in PLGA microspheres 273 Encapsulation 6, 7, 8, 12, 22, 58, 80, 100, 157, 158, 168, 202, 210, 221, 259 density 100 efficiency 7, 8, 12 Endocytosis 38, 70, 132, 135, 181 cellular 181 process 70 Endometriosis 277 Enzymes 61, 101, 102, 113, 138, 153, 179, 180, 215, 224, 252 copper-binding protein 215 nitrate reductase 138 proteolytic 101, 224 purified CGTase 180 Epichlorohydrin 173 European medicines agency (EMEA) 168

#### F

Factors 23, 29, 33, 136, 177, 179, 213, 215, 220, 246, 251, 253, 255, 264, 269 angiogenic 269 genetic 213 inflammatory 264 inflammatory physicochemical 255 transcriptional 136 Field-effect transistor (FET) 217 Fine particle fraction (FPF) 270 Flow cytometry 225 Fluorescent silica 79, 80, 81 based nanoparticles 80, 81 nanoparticles 79 Foam casting technique 94 Food and drug administration (FDA) 94, 169, 204, 208, 212, 223, 227, 228, 268, 275

## G

GelsConstruct preparation technique 260

Nanomaterials: Evolution and Advancement (Part 1) 297

#### Η

Helicobacter pylori infection 214 Hepatic 141 carcinoma 141 glycogenesis 141 Hepatocellular cancer/solid tumors 227 Hereditary liver disorders 227 Higher 94, 108 alkaline phosphatase 94 proteoglycan synthesis 108 High pressure homogenization (HPH) 16, 17, 30 HIV 216, 219 infection 216 reverse transcriptase enzyme 219 HIV-1 216 encephalitis 216 associated neurocognitive disorders 216 Hormones 222, 273 inhibited luteinizing 273 thyroid-stimulating 222 Horseradish peroxidase (HRP) 113, 117 Human 111, 203, 204, 208, 219, 228 adipose stem cells 111 brain carcinomas 204 colon cancer xenograft 208 immunodeficiency virus (HIV) 203, 219, 228 umbilical vein endothelial cells (HUVECs) 111 Hydrogel for bone tissue repair 93, 98 Hydrolysis reaction 65 Hydrophile-lipophile balance (HLB) 9, 10, 15 Hydrophilic drug release 114 Hydrophobic ion pairing (HIP) 16, 21, 23, 39, 136 Hydrophobicity 8, 62, 154, 155, 246, 252 Hydrophobic 21, 165 lipid matrix 21 properties 165 HyperBranch medical technology 221

298 Nanomaterials: Evolution and Advancement (Part 1)

#### Nimesh et al.

## Ι

Imaging of cardiovascular diseases 217 Immune response 101, 135, 223, 224, 264, 265 Immunoassays 78, 138, 140 Implant preparation techniques 260 Inductively coupled plasma optical emission spectrometry (ICPOES) 70 Infections 95, 99, 101, 129, 139, 218, 219, 222, 223, 247, 255, 256, 257, 269 antibiotic-resistant 218 catheter-related bloodstream 139 chronic wound 95, 99 gastric 218 multiple pulmonary 223 Inflammation 108, 221, 222, 226, 247, 255, 256, 257, 269, 272 chronic 255, 256 Inflammatory demyelination 214 Injury 106, 110, 129 endometrial 106 traumatic brain 110 Insulin 96, 99, 133, 141, 199, 270, 272, 273 cyclodextrin 273 formulations 273 mediated 133 protected 96 Interactions 28, 29, 72, 90, 91, 96, 107, 110, 151, 152, 154, 155, 156, 158, 161, 163, 164, 165, 174, 176, 204, 215 cell hydrogel 96 drug-polymer 151, 152 electrochemical 215 electrostatic 72, 110, 204 Irritable bowel syndrome 114 Ischemic heart disease 276

## K

Kidney diseases 181 diabetic 181 *Klebsiella pneumonia* 179 Kneading technique 156

## L

Laser 24. 63 diffraction (LD) 24 doppler anemometry 63 Lateral sclerosis 215 Leishmaniosis 224 Leukemia, chronic lymphocytic 212, 228 Leukemia inhibitory factor (LIF) 214 Lipid 5, 12, 21, 22, 40 crystallization rate 40 drug conjugates (LDC) 21, 22 hydrophilicity 12 hydrophobicity 12 matrix organization 5 Liposome-encapsulated hemoglobin (LEH) 216 Listeria monocytogenes 224 Listeriosis 224 Lou Gehrig's Disease 215 Low-density lipoproteins (LDL) 203, 217 Luciferase assay 262 Lung 37, 209, 226, 274 cancer 37, 209, 226 disease 274 metastasis 37

## Μ

Mass spectrometry sorptometer 63 Maximum tolerable dose (MTD) 202, 209, 226 Measles 129, 223 Mechanism 133, 134, 136, 138, 141, 151, 160, 164, 249, 258, 262, 264, 265 cross-primping 264 tumor necrosis factor antagonist 133 Membrane contactor method 19 Mesoporous silica nanoparticles (MSNs) 57, 58, 59, 60, 61, 63, 64, 69, 70, 71, 72, 73, 74, 75, 76, 77, 78, 79, 80, 206 Metabolic 133, 209, 210, 227, 228 diseases 133 breast cancer 209, 210, 228

#### Subject Index

pancreatic cancer 227 Methods 67, 28, 68, 248, 261 chemical vapour condensation 68 depolymerization 248 high-performance liquid chromatography 28 melt polycondensation 248 scanning electron microscopy 261 spray pyrolysis 67 Micrococcus luteus 134, 179 Microparticle preparation techniques 258 Monoclonal antibodies 152, 197, 214 Monocytes 254, 255 inflammatory 255 MRI imaging 76 MRSA infections 218 MSNs for biomedical applications 74 Multimodal imaging methods 80 Multiple sclerosis (MS) 129, 212, 214, 215, 228 Myocardial 111, 217, 276 infarction (MI) 217 ischemia 111, 217, 276

## Ν

Nanoparticle-based 205, 217, 218, 223, 224 cardiovascular imaging 217 delivery of anti-infective agents 218 therapeutics 205 vaccines 223, 224 Nanoparticles preparation techniques 259 Nanoprecipitation method for anticancer therapy 162 Nasal vaccination 224 Negligible cellular reaction 257 Nektar therapeutics 209 Neuroblastoma cells 216 Neuroinflammation 213 Neuronal death 212 Neurons 213, 215 dopaminergic 213 Neuroprotectors 203 Neutropenia 208 febrile 208

Nanomaterials: Evolution and Advancement (Part 1) 299

NMR 105, 154, 250 spectra 154 spectroscopy 250 Non-small cell lung cancer (NSCLC) 208, 209, 210, 226, 227 NSCLC and metastatic breast cancer 209 Nuclear magnetic resonance techniques 30 Nucleic acid detection assays 139

## 0

Objective response rate (ORR) 207 Octapeptide based hydrogel 114 Optical coherent tomography 217 Osteoarthritis 204 Osteogenesis 93 Osteoporosis 275 Oxidation hydrophilicity 60

## P

Pancreatic cancer 143, 167, 207, 208, 227 progressed metastatic 207 Parkinson's disease (PD) 194, 207, 212, 213, 227, 229 Peptides 77, 90, 91, 92, 94, 95, 99, 101, 102, 109, 110, 111, 112, 113, 114, 116, 216, 222, 258, 269 antigenic 102 cleaving hydrogel linker 101 cyclic 101 mitogenic 269 pH-responsive Hydrogel Facilitating 102 therapeutic 77, 99 trans-activating transcriptor 216 vasoactive intestinal 222 Peripheral blood mononuclear cells (PBMC) 203 Phagocytosis 254, 264 Photon correlation spectroscopy (PCS) 24, 62 Photothermal Synthesis 68 Piezoresponse force microscopic (PFM) 103 Pike technologies 261

#### 300 Nanomaterials: Evolution and Advancement (Part 1)

Plasma proteins 159 Plasmid DNA 73, 202, 210, 261, 265, 267 freeze-dried 261 Plasmon resonant particles 81 PLGA 246, 247, 251, 252, 253, 266, 267, 272, 273, 276 and copolymers for drug delivery applications 276 biodegradation of 246, 252 degradation of 251, 252, 253 mechanical properties of 252 microparticles 247, 266, 272, 273 mPEG nanoparticles for prostate cancer 267 PEG blend formulation 273 synthesis 247 PLGA copolymers 250, 251, 252, 253 possessing 253 PLGA microcapsules 269, 273 biodegradable 269 Polycondensation reactions 247 Polylysine-based hydrogel 106 Polymeric hydrogels 91, 94, 98, 99, 102, 105, 107 pH-responsive 102 reported pH-dependent 105 Polymeric nanocarriers 3, 168 drug-loaded 168 Polymerization 173, 198, 249, 250 interfacial 198 process 250 reaction 249 Polymers 14, 16, 35, 90, 91, 92, 96, 99, 105, 166, 169, 197, 198, 213, 219, 247, 248, 249, 250, 251, 252, 253, 254, 258, 260, 262, 264, 270, 278 biocompatible 264 coacervate phase 258 crystalline 251, 252 ethylene oxide 14 hydrophilic 16, 35 hyper-branched 169 protein 99 thermosensitive tetronic 96 toxic 213 Polyvalent inhibitors 133

Positron emission tomography (PET) 76, 144, 206, 217 Prions disease 216 Process 61, 67, 68, 77, 97, 112, 128, 132, 133, 137, 138, 139, 158, 173, 177, 178, 179, 180, 214, 215, 246, 258, 260, 261, 264 cyclodextrin production 179 degenerative 214, 215 drug complexation 173 dynamic 158 fast cooling 68 gas foaming 261 heating 128 hydrogelation 112 modified Tollens 137 multiple-step healing 139 oil-in-water emulsification 246 regenerative 97 sol-gel synthesis 77 Progression-free survival (PFS) 207 Prostate cancer 167, 210, 226, 267, 276, 277 Prostatic carcinoma 167 Protease catalysis 112 Proteins 77, 78, 105, 107, 110, 112, 115, 127, 128, 130, 196, 198, 212, 214, 216, 225, 257, 258, 259, 265, 267 anti-angiogenic 112 bovine serum albumin 130 elastin 110 keratin 115 myelin 214 prion 216 tau 212

## Q

Quality of life (QoL) 195, 212, 214

## R

Reaction 11, 67, 68, 93, 101, 105, 113, 130, 134, 157, 179, 248, 225, 226, 250, 254, 255, 256 allergic 225

#### Nimesh et al.

#### Subject Index

enzyme 179 esterification 105 hypersensitivity 203 immune 115 inflammatory 93, 101, 254 lipolysis 11 osteolytic 255 pathogenic 226 Reactive oxygen species (ROS) 141, 142, 143, 203, 212, 215 Reperfusion ischemia 203

## S

Scaffolds 33, 94, 107, 246, 260, 261, 268, 269 engineering 94, 107 preparation technique 261 Scanning 24, 25, 27, 61, 109, 174 electron microscopy 27, 61, 109, 174 ion occlusion sensing (SIOS) 24, 25 Shrinking tumor load 220 Sickle cell anemia 172 Silica nanoparticles 77, 79, 80, 81 antimicrobial mesoporous 80 ester-functionalized polypyrrole 77 luminescent 77 magnetic mesoporous 79 polymer-coated mesoporous 81 Silver 112, 128, 129, 131, 136, 138, 139, 140, 144, 219, 221 metallic 221 nano-crystalline 139 reducing ionic 112 Single crystal X-ray diffraction 175 technique 175 Single-photon emission computerized tomography (SPECT) 76, 217 Skin 32, 33, 35, 91, 129, 168, 172, 198, 204, 218, 219, 257 cancer 168 subcutaneous 257 ulcer 129 Small 208, 209, 212, 227, 228 cell lung cancer (SCLC) 208, 209, 227

#### Nanomaterials: Evolution and Advancement (Part 1) 301

lymphocytic lymphoma (SLL) 212, 228 Solvent evaporation techniques 67, 259 Solvent injection method 18 Spectroscopy 24, 25, 62, 93, 95, 110, 176 energy-dispersive X-ray 93 photon correlation 24, 62 scanning ion occlusion 25 Spray-drying 157, 259, 271 method 157 techniques 259, 271 Squamous cell carcinoma 210 Stability 8, 11, 25, 26, 62, 64, 66, 90, 96, 137, 138, 160, 163, 166, 196, 271, 273 chemical 8, 196 mechanical 90 thermodynamic 66 Staphylococcus aureus 134, 218 Stargardt's diseases 181 Starpharma for bowel cancer 210 Stem cell 93, 98, 108, 110, 111 culturing bone marrow 93 human bone marrow 98 implanted neural 110 mesenchymal 108 Sterically stabilized micelles (SSM) 222 Stober process 64, 65 Stokes-Einstein equation 103 Stratum corneum (SC) 33 Stress 17, 30, 59, 106, 110, 142, 203, 215, 226 mechanical 59 oxidative 142, 203, 215, 226 shear 17 Summation of common imaging techniques 144 Supramolecular systems 166 Surface-enhanced raman scattering (SERS) 138 Surfactants 8, 11, 12, 13, 14, 15, 18, 20, 22, 30, 38, 40, 64, 137, 196 amphoteric 14 cytotoxic 8 Symptoms 170, 212, 216, 225, 255 associated neurologic 212 psychiatric 212 toxicological 170

#### 302 Nanomaterials: Evolution and Advancement (Part 1)

Synthesis 59, 64, 67, 91, 94, 125, 130, 131, 132, 135, 137, 138, 174, 180, 247, 248, 249 aerosol decomposition 67 dual enzyme 180 enzymatic 174 extracellular biological 131 Systems 2, 3, 12, 25, 26, 29, 30, 62, 64, 73, 74, 98, 99, 106, 114, 116, 177 alginate-based hydrogel 98 antibacterial release 99 cryopreservative polymeric ampholyte 106 delayed-release drug carrier 177 engineered protein-based polymeric hydrogel 99

## Т

Target cancer 116, 208 cells 208 therapy 116 Targeted delivery 194, 225 in nanotechnology 194 of nanotherapeutics 225 Targeting 40, 200, 212, 220 cancerous cells 40 CNS disorders 212 hereditary liver disorders 200 intraocular carcinomas 220 Techniques 16, 17, 18, 19, 23, 24, 26, 27, 58, 61, 62, 76, 103, 133, 157, 174, 176, 258, 259, 260, 262 electrospinning 262 microparticles 259 of emulsification 259 of microparticles preparation 259 oil-in-water 258 spectroscopic 103, 174 synthesized 58 therapeutic 133 ultrasonication 17 Technology 19, 116, 133, 198 hybrid imaging 133 recombinant DNA 116

supercritical fluid 19, 198 Tetanus toxoid (TT) 224 Therapy 40, 71, 72, 73, 74, 80, 100, 101, 125,132, 133, 169, 200, 211, 229, 266 adoptive cell 101 anti-infective 200 antisense 73 hormone replacement 169 photodynamic 80, 132 Thermal 18, 175 analytical technique 175 stress 18 Thermo derivatography 175 Thyroid-stimulating hormone (TSH) 222 Tobacco mosaic virus (TMV) 224 Tollens method 137 Tomography 76, 133, 144, 217 computed 217 computerized 144 photoacoustic 133, 217 Transcytosis 38 Tuberculosis 219 Tumor cells 142, 143, 203, 205, 209, 210, 211, 266, 267 invasive epithelial breast 267 oxygen-deprived 205 papillary thyroid 211 Tumors 37, 76, 128, 168, 207, 208, 211, 222, 227.266 effective 211 epithelioid cell 208 neuroendocrine 207, 227 targeting malignant 266 Turkevich method 130

#### U

Uterus, injured 106

## V

Vaccination 92, 101, 107, 114, 117, 224, 264, 265 adjuvant in 114, 117 oral 101

#### Nimesh et al.

#### Subject Index

Vaccines 102, 107, 114, 194, 197, 201, 223, 224, 264, 265, 266 effective 264, 265 live attenuated 114 mucosal 224 nanoparticles-based 102 non-encapsulated recombinant 224 ovananoparticle 223 siRNA-based 223 Vascular 111, 269, 276



## Surendra Nimesh

Dr. Surendra Nimesh is an internationally recognized nanotechnology expert for biological applications with a specialization in drug and gene delivery. He obtained his MSc degree in Biomedical Science and a Ph.D. degree in Nanotechnology from Dr. B.R. Ambedkar Center for Biomedical Research, University of Delhi and CSIR- Institute of Genomics and Integrative Biology, Delhi, respectively. He is currently working as a UGC assistant professor in the Department of Biotechnology at the Central University of Rajasthan, India. He has authored several research and review articles in international peer-reviewed journals, book chapters, and 4 books. He worked as a postdoctoral fellow at the Ecole Polytechnique of Montreal, the Clinical Research Institute of Montreal, the McGill University, and a visiting fellow at the Health Canada, Canada, during 2007- 2013. His area of interest includes nanotechnology for drug/gene/siRNA delivery for the treatment of cardiovascular diseases and breast cancer.



Nidhi Gupta

Dr. Nidhi Gupta received her MSc degree in Biomedical Science from Dr. B.R. Ambedkar Center for Biomedical Research, University of Delhi, in 2004 and Ph.D. in Virology from the National Institute of Immunology (NII), JNU, India, in 2008. She worked as a postdoctoral fellow at the Clinical Research Institute of Montreal (2007-2008) and the McGill University (2011-2013), Montreal, Canada. She has successfully completed the DST Young Scientist project at the IIS University, India, sponsored by SERB, India. Currently, she is working as a senior research scientist at the IIS University, Jaipur. Her field of interest includes catalytic nucleic acids, siRNA, and nanotechnology for drug delivery.



## **Ramesh Chandra**

Prof. Ramesh Chandra is a distinguished scientist, a fellow of the Royal Society of Chemistry, London, and an outstanding Biomedical Sciences researcher. He is the founder-director of Dr. B. R. Ambedkar Center for Biomedical Research, University of Delhi. Professor Chandra started his research career at the University of Delhi; thereafter, he went to The New York Hospital-Cornell University Medical Center and the Rockefeller University, New York; the State University of New York at Stonybrook as an assistant research professor. He conducted advanced research at the Harvard University Medical School-Massachusetts General Hospital, jointly at MIT, Cambridge, USA. Over the last 39 years, professor Chandra has contributed largely in the field of chemical sciences and particularly in new drug discovery and development as well as drug metabolism. He has published more than 400 research papers in International journals of repute, including review articles/monographs, book chapters; Four of his chapters are also published in internationally acclaimed books.